Dear Editor,

We would like to thank you for accepting to reconsider our manuscript titled: “Graft choices for anterolateral ligament knee reconstruction surgery: Current concepts” for publication in the World Journal of Clinical Cases. We would also like to thank the reviewers for their insightful comments. All points raised are addressed and the manuscript was revised according to their suggestions. For reviewing purposes, the comments have been addressed one by one and all text changes in the manuscript have been highlighted.

In more detail:

**Reviewer #2:**

**Comment:** The article is within the scope of the journal and deals with an interesting topic. It is well written and easy to read. However, it is necessary to make some improvements to be accepted: a) It should be better explained what type of systematic or narrative review, indicate the bibliographic sources, how the search has been carried out, and how the results will be grouped to present them. b) At the end of the article there should be a discussion section where the conclusions of the review carried out are established and it is indicated how future advances in this area can progress. c) In the conclusions they should also add a set of lines of future work regarding how to improve the review carried out.

**Response:** Thank you for the comments. a) This article is a narrative review that contacted using the MEDLINE online database and the available bibliographic sources regarding ALLR are presented in the text and grouped in Table 1 according to the available graft types, graft fixation techniques and final functional outcomes. b) A Discussion section has been added presenting the main conclusions of the review and in which fields future research should
be carried out to clarify the role of ALLR in the outcome of ACLR surgery. e) The Conclusion section has been modified accordingly to point out how the current review can be further improved.

Once again, we would like to thank you and the reviewers for your time and effort. Please do not hesitate to contact me for any further clarifications and corrections regarding the submitted Manuscript.

Yours sincerely,
Byron Chalidis, MD, PhD
Corresponding Author